Author's response to reviews

Title: Childhood tuberculosis desk guide and monitoring: An intervention to improve case management in Pakistan

Authors:

Nauman Safdar (safdar.nauman@gmail.com)
Sven Gudmund Hinderaker (Sven.Hinderaker@cih.uib.no)
Noor Ahmed Baloch (noorahmad14@hotmail.com)
Donald A Enarson (denarson@theunion.org)
Muhammad Amir Khan (asd@asd.com.pk)
Odd Morkve (Odd.Morkve@cih.uib.no)

Version: 4 Date: 9 August 2011

Author's response to reviews: see over
**Cover letter:**

This is in reference to our submitted manuscript in BMC health services research entitled “Childhood tuberculosis deskguide and monitoring: An intervention to improve case management in Pakistan”.

It was advised by the editor to submit the manuscript, after suggested formatting.

1. Authors affiliations: Departments and addresses are edited and are as per the suggested format
2. Table format: The tables are formatted

We are submitting the revised manuscript.